Author’s response to reviews

Title: Predicting progression of mild cognitive impairment to dementia using neuropsychological data: a supervised learning approach using time windows

Authors:
Telma Pereira (telma.lm.pereira@gmail.com; telma.pereira@ist.utl.pt)
Luís Lemos (luismatiaslemos@gmail.com)
Sandra Cardoso (sandradcardoso@gmail.com)
Dina Silva (dlsilva@ualg.pt)
Ana Rodrigues (ana.pina.rodrigues@gmail.com)
Isabel Santana (isabeljsantana@gmail.com)
Alexandre de Mendonça (mendonca@medicina.ulisboa.pt)
Manuela Guerreiro (mmgguerreiro@gmail.com)
Sara C. Madeira (sara.madeira@tecnico.ulisboa.pt)

Version: 1 Date: 10 Mar 2017

Author’s response to reviews:

Dear Editor,

Please consider our resubmission of manuscript MIDM-D-17-00055 "Predicting the progression of mild cognitive impairment to dementia using neuropsychological data: a supervised learning approach using time windows".

As requested, we clarified the source of the data used in the study as well as the ethics approval statement. In particular, we completed the ethics approval and consent to participate section with the following text: "The study was conducted in accordance with the Declaration of Helsinki, and was approved by the local (Faculty of Medicine, University of Lisbon) ethics committee. Informed consent to participate in the study was obtained from all participants. Data access was
granted in the context of project NEUROCLINOMICS2 (PTDC/EEI-SII/1937/2014) where the author’s institutions participate." Moreover, we completed the Data section of the manuscript with the following text "The study was conducted in accordance with the Declaration of Helsinki, and was approved by the local ethics committee. Informed consent to participate in the study was obtained from all participants”.

Sincerely,

Telma Pereira on behalf of all the authors